

Article



The Interplay Between Environmental Ethics and Sustainable Performance: Does Organizational Green Culture and Green Innovation Really Matter?

Elhassan Enbaia, Ahmad Alzubi *, Kolawole Iyiola and Hasan Yousef Aljuhmani

Department of Business Administration, Institute of Graduate Research and Studies, University of Mediterranean Karpasia, Mersin 33010, Turkey; kolawole.iyiola@akun.edu.tr (K.I.) * Correspondence: ahmad.alzaubi@akun.edu.tr

Abstract: Although organizational green culture and environmental ethics are increasingly acknowledged as crucial, few studies have comprehensively examined the interconnected relationship between green innovation and sustainable performance, particularly within manufacturing firms. As such, this study empirically examines the relationship between environmental ethics and sustainable performance through the mediating roles of green process and product innovations and the moderating role of organizational green culture on these relationships. It is grounded in stakeholder theory and a resource-based view. This study utilizes 236 valid cross-sectional data points from manufacturing firms in Turkey to test the proposed integrated theoretical model through structural equation modeling (SEM). Environmental ethics is positively associated with sustainable performance, green process innovation, and green product innovation. Green process innovation positively affects sustainable performance. Additionally, both green process and product innovations mediate the relationship between environmental ethics and sustainable performance. Moreover, organizational green culture enhances the relationship between environmental ethics and both green process innovation and green product innovation while moderating its role in sustainable performance. The findings highlight how environmental ethics, mediated by green innovation and moderated by organizational green culture, can drive sustainable performance, offering valuable insights for managers aiming to enhance sustainability initiatives in the manufacturing sector.

Keywords: environmental ethics; green process innovation; green product innovation; organizational green culture; sustainable performance; Turkey's manufacturing sector

1. Introduction

Organizations are increasingly embedding sustainability into their strategies and operations owing to heightened public awareness of environmental issues [1,2]. The manufacturing sector, which is essential for global economic growth and job creation, is one of the largest contributors to environmental challenges such as resource depletion and ecological degradation [3,4]. With rising discomfort from global warming and the need for ecological balance, stakeholders are placing greater pressure on manufacturing firms to adopt sustainable performance and decrease the environmental impact of their production processes [5]. Despite the acknowledged importance of sustainability, the link between environmental ethics, green innovation, and sustainable performance remains underexplored in the academic literature. While more studies have recognized that green human resource management and CSR influence green innovation and sustainability, the role of environmental ethics, particularly in driving green process and product innovations, has not been extensively examined [6,7]. This leaves a significant gap in

Citation: Enbaia, E.; Alzubi, A.; Iyiola, K.; Aljuhmani, H.Y. The Interplay Between Environmental Ethics and Sustainable Performance: Does Organizational Green Culture and Green Innovation Really Matter? *Sustainability* **2024**, *16*, 10230. https://doi.org/10.3390/su162310230

Academic Editors: Paola Cocca and Joseph Paul Zammit

Received: 26 September 2024 Revised: 19 November 2024 Accepted: 20 November 2024 Published: 22 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/license s/by/4.0/). understanding how environmental ethics, when integrated into an organization's culture, can contribute to long-term sustainability [8].

This study seeks to fill this gap by exploring the relationship between environmental ethics, green innovation, and sustainable performance within the context of manufacturing firms in Turkey. In this study, environmental ethics refers to the ethical principles that guide a firm's responsibility toward environmental protection. The study also introduced organizational green culture as a moderating factor in the link between environmental ethics and sustainable performance [9]. The focus on Turkey's manufacturing sector provides a novel context, as environmental sustainability in emerging markets has been less studied than in developed economies. Sustainability initiatives and environmental ethics are widely recognized as contributing to enhanced organizational performance, particularly in the context of green innovation. An empirical study demonstrated that firms that adopt green innovations tend to achieve better financial and environmental outcomes [10]. However, the specific pathways through which environmental ethics drives green innovation and sustainable performance, especially within the manufacturing sectors in emerging markets, remain poorly understood.

The function of environmental ethics as a supporter of green innovation and sustainable performance has not been studied extensively, particularly in the manufacturing sector. Additionally, empirical studies exploring the mediating role of green innovation and the moderating role of organizational green culture in this relationship are scarce. This study seeks to address these gaps by investigating how environmental ethics is associated with both green process and product innovations and how these innovations, in turn, relate to sustainable performance [11]. Moreover, this study investigates how a firm's green culture may strengthen or weaken these relationships.

This research aims to concentrate on the following research questions:

RQ1: What is the nature of the relationship between environmental ethics and sustainable performance among manufacturing firms?

RQ2: What is the nature of the relationship between environmental ethics and green innovation?

RQ3: How do green process and product innovations mediate the relationship between environmental ethics and sustainability performance?

RQ4: How does organizational green culture moderate the relationships among environmental ethics, green innovation, and sustainable performance?

This study contributes to the literature by addressing the gap in empirical evidence concerning the role of environmental ethics in promoting green innovation and sustainable performance. It also highlights the importance of organizational green culture as a critical moderating factor in achieving these outcomes. By offering new insights into the interplay between environmental ethics, green innovation, and sustainability in the manufacturing sector, this study makes both theoretical and practical contributions. Managers can benefit from understanding how fostering a green organizational culture can enhance their firm's sustainability initiatives and overall competitiveness.

The remainder of the paper is organized as follows: Section 2 reviews the relevant literature, presents the conceptual framework, and develops hypotheses related to the direct and indirect relationships through which environmental ethics are associated with sustainable performance. Section 3 outlines the research methodology, including data collection and sampling techniques. Section 4 presents the results, and Section 5 discusses the findings and concludes with implications, limitations, and suggestions for future research.

2. Theoretical Background and Hypothesis Development

2.1. Stakeholder Theory

Stakeholder theory, introduced by Freeman [12], emphasizes the role of stakeholders and environmental organizations in driving businesses to adopt social and environmental

3 of 24

responsibilities. Effective stakeholder engagement leads to sustainable business practices, including green innovation and the development of an organizational green culture [13]. These relationships are built on cooperation, consultation, participation, and information sharing, thus enhancing sustainability efforts [14].

This study applies stakeholder theory to examine the link between green practices and sustainable performance in Turkey's manufacturing sector. Organizations with strong stakeholder theory are achieving social sustainability, aligning with Turkey's work ethics. Moreover, stakeholder theory advocates fairness, honesty, and adherence to ethical codes in all stakeholder interactions, which improves sustainability performance and benefits both the business and its surrounding community [15].

2.2. Resource-Based View

Hart's resource-based view theory offers a nuanced understanding of how organizations interact with their environment, addressing the limitations of traditional resourcebased view frameworks [16]. This perspective emphasizes that an organization's interaction with natural resources can enhance its competitive advantage by fostering environmentally sustainable practices. The resource-based view highlights the necessity for organizations to develop expertise in environmental stewardship, recognizing the profound influence of ecological factors in today's competitive landscape. The authors of [17] further argue that, to sustain competitive advantage in the future, businesses must engage in environmentally friendly economic activities, making this commitment central to their operational strategy.

By cultivating a green culture and adopting ethical practices, organizations can enhance productivity and reduce operational costs through innovations in green processes and products. This alignment with resource-based view principles not only leads to improved financial performance but also positions firms favorably in the marketplace [13]. In this study, the resource-based view was used to highlight the importance of green processes and products at the organizational level. This research focuses on how effective interactions with the environment contribute to sustainable performance, ultimately resulting in increased productivity and decreased operational expenses. The accomplishment of green processes and products fosters enhanced environmental security, social performance, and competitive advantage [18].

2.3. Environmental Ethics and Sustainable Performance

Environmental ethics serves as a crucial internal resource that enables firms to develop value-creating strategies, ultimately enhancing their sustainable performance. Sustainable performance refers to how organizations engage with the natural environment, emphasizing resource consumption and stringent pollution control measures [19]. Utilizing the resource-based view, it is posited that firms can naturally formulate strategies that influence their internal resources, including environmental ethics, to improve sustainable performance and achieve a competitive advantage [16]. Moreover, a firm's commitment to corporate environmental ethics fosters proactive environmental initiatives that positively influence its sustainable performance. Such initiatives may include the use of biodegradable materials in products, reducing pollution and waste at the source, minimizing the use of environmentally harmful substances, and enhancing energy efficiency [20]. This proactive approach not only enhances an organization's social acceptability but also provides a competitive edge in the marketplace.

Therefore, organizations are encouraged to adopt environmental management as a core organizational philosophy, involving all employees in the greening process [21]. A substantial body of research supports the addition of environmental ethics to firms' planned methods and organizational green culture to promote sustainable performance [22]. Drawing on the resource-based view, it has been asserted that corporate environmental ethics, as a valuable and rare resource, fosters an environmental philosophy that

governs the management of people, processes, and products, thereby enhancing sustainable performance [16,18]. Thus, we propose the following hypothesis:

H1. *Environmental ethics is positively associated with sustainable performance.*

2.4. Environmental Ethics and Green Process Innovation

Green innovation aims to achieve corporate environmental management [23]. This study categorizes green innovation into two distinct types: green product and process innovations. By improving environmental management performance, green innovation enables companies to meet their environmental protection standards. Firms investing in green innovation not only comply with environmental regulations but also establish competitive barriers against rivals. Historically, many organizations have viewed environmental management as an unnecessary expense [24]. However, an increasingly positive correlation exists between corporate environmental ethics and green process innovation. Organizations are urged to realign their strategies and operations to adapt to the growing emphasis on environmental sustainability.

Companies that demonstrate strong environmental ethics typically enhance resource productivity through green innovation, effectively mitigating environmental costs [25]. Green innovation not only enhances product value but also offsets the costs associated with minimizing environmental impact, thereby increasing overall competitiveness. A study employing content analysis found that green process innovation significantly strengthens a firm's organizational environmental ethics [26]. Furthermore, CSR has been shown to significantly influence firms' adoption of green process innovation. Similarly, stakeholder force plays a decisive role in cheering on associations to implement green process innovation [27]. Based on this discussion, we propose the following hypothesis:

H2. Environmental ethics is positively associated with green process innovation.

2.5. Green Process Innovation and Sustainable Performance

Climate change presents significant environmental challenges that impede organizational sustainability. To combat these escalating ecological issues, proactive strategies are essential to achieve long-term performance [28]. Within the framework of sustainable development, green product innovation has emerged as a strategic approach that meets stakeholder needs while providing firms with a lasting competitive advantage. These innovations encompass a range of unique business processes that enhance a firm's sustainable performance [29]. The substantial impact of green product innovation highlights the necessity of implementing eco-friendly practices as a critical driver of organizational sustainability [30].

Numerous studies have identified green process innovation as a vital driver of sustainable performance [31]. In today's industrial landscape, green modifications have significantly improved processes and systems, reduced environmental vulnerabilities, and promoted sustainable production. Manufacturing organizations that adopt green practices experience notable improvements in their sustainable performance [32]. Given its substantial contribution to economic development, the manufacturing sector must prioritize green product innovation to enhance profitability and overall performance [33]. Research further demonstrates that green product innovation is highly beneficial for firms seeking sustainable development and plays a pivotal role in improving social, economic, and ecological outcomes [34]. Based on this argument, we propose the following hypothesis:

H3. *Green process innovation is positively associated with sustainable performance.*

2.6. Environmental Ethics and Green Product Innovation

Green product innovation highlights products that reduce their environmental impact throughout their life cycles [35]. By focusing on green products, companies can conserve resources, mitigate environmental risks, and reduce waste. Improving product design through green innovation enables firms to achieve their environmental objectives. In this study, green product innovation refers to the extent to which companies develop or improve products to mitigate negative environmental effects [36]. To facilitate this, organizations must possess both the motivation and capability to generate innovative ideas for new product development. Previous research highlights the positive impact of well-defined environmental policies and processes on a company's innovation efforts [37]. Clear environmental policies can enhance operational integration across departments and address environmental challenges effectively.

Corporate environmental ethics plays a critical role in proactive environmental management by influencing the adoption of environmental technologies and overall business operations. This study posits that environmental ethics significantly contributes to a company's green innovation, with the breadth and speed of a firm's response to environmental issues positively correlated with top management's environmental concerns [38]. According to the resource-based view, a strong corporate culture that is characterized as valuable, rare, inimitable, and non-substitutable can serve as a vital resource for sustained competitive advantage [18]. Environmental ethics, as a form of superior corporate culture, significantly contributes to sustainable development. Thus, companies with robust environmental ethics employ positive environmental measures to facilitate green innovation [16]. Based on this discussion, we propose the following hypothesis:

H4. Environmental ethics is positively associated with green product innovation.

2.7. Green Product Innovation and Sustainable Performance

As globalization accelerates, businesses are confronted with resource limitations and ecological degradation, significantly hindering their sustainable development. Environmental deterioration is a critical challenge across industries worldwide, necessitating proactive and innovative eco-friendly strategies to combat climate change. In this study, environmental innovation and green attributes have become central to achieving sustainability. Research indicates that green product innovation is essential to maintaining a company's long-term sustainability [39]. Furthermore, it plays a pivotal role in reducing the escalating environmental impact on businesses [40]. Consequently, prior studies have advocated the integration of green practices to enhance sustainable performance [41].

Product innovation now serve as a critical driver for accelerating growth within firms as the contemporary business landscape increasingly encourages the adoption of green innovation tools to achieve sustainable performance. The literature highlights a positive relationship between green product innovation and sustainable performance. As organizations increasingly embrace eco-friendly practices, they can effectively mitigate the adverse effects of climate change. Green product innovation, which is characterized by environmentally conscious products, provides firms with a competitive advantage over their market rivals. Stakeholders can leverage these innovations to enhance an organization's sustainability in a highly competitive environment [33]. Based on this discussion, we propose the following hypothesis:

H5. *Green product innovation is positively associated with sustainable performance.*

2.8. The Mediating Role of Green Process Innovation

Environmental ethics management entails strict adherence to all relevant laws, regulations, and environmental requirements, enabling companies to minimize their ecological impact while ensuring compliance with quality standards [42]. Research highlights the critical role of SMEs in implementing environmental strategies that enhance organizational effectiveness and improve environmental performance [43]. Stakeholders and public authorities expect SMEs to exhibit both environmental and financial performance. This expectation is particularly pronounced for SMEs as they depend more on local communities and engage in closer interactions than larger corporations [44]. Thus, stakeholder-focused ecological fulfillment and disaster improvement are necessary for endurance.

Previous research has established that green process innovation contributes positively to both environmental and economic performance. Moreover, studies indicate that green process innovation enhances a company's competitive advantage and overall sustainability [45]. A significant relationship has also been identified between green process innovation and a firm's reputation and environmental ethics, particularly within the context of SMEs [46]. Additionally, eco-efficiency is vital for achieving environmental performance, with green process innovation serving as a key factor in enhancing sustainability performance, where environmental ethics acts as a mediating variable [47]. Based on this discussion, we propose the following hypothesis:

H6. *Green process innovation mediates the relationship between environmental ethics and sustainable performance.*

2.9. The Mediating Role of Green Product Innovation

Green product innovation performance refers to the growth of narratives and useful ideas related to green products, services, processes, or practices [48]. For companies seeking to create and market green products effectively, it is essential to integrate green management principles with green product innovation activities. According to [49], green innovations encompass both hardware and software innovations associated with green products. Their research indicates that investing in green product innovation positively impacts sustainable performance, underscoring the value of recognizing the role of green innovation in business strategy.

As green products become increasingly prevalent globally, the significance of green product innovation is expected to grow. Heightened public awareness of environmental issues has also contributed to a growing focus on green product innovation. Companies design green products to convert market opportunities into offerings that meet the needs of consumers and other stakeholders. However, integrating environmental considerations into product development while complying with regulations and market demands can be challenging [50]. This challenge is underscored by the need for a systematic and innovative approach to designing and innovating products that diminish ecological impact. Growth of ecologically friendly products to achieve company success [51]. Therefore, we propose the following hypotheses:

H7. Green product innovation mediates the relationship between environmental ethics and sustainable performance.

2.10. The Moderating Effect of Organizational Green Culture

Organizational green culture is pivotal in shaping employees' mental frameworks and guiding their behaviors. It significantly influences innovation activities within organizations, either by fostering or obstructing them [52]. Among the various factors driving green practices, organizational green culture is particularly noteworthy because establishing a robust green organizational culture is essential for promoting successful green innovation. Scholars suggest that such a culture serves as an optimal framework for achieving success in green innovation initiatives, as embedding environmental awareness into the organizational fabric is vital for stimulating green innovation [53]. The resource-based view framework elucidates the relationship between organizational green culture and both green product and process innovations, emphasizing the strategic importance of leveraging internal resources to attain and maintain a competitive edge [16]. In this research, businesses increasingly view organizational green culture as a valuable and unique asset for achieving sustainable performance. Research indicates that organizational green culture significantly impacts green product and process innovations across both the service and manufacturing sectors, acting as a catalyst for green innovation and sustainable performance [54]. Furthermore, green innovation mediates the relationship between organizational green culture and sustainable performance. Several studies have substantiated the positive correlation between organizational green culture and green innovation, emphasizing that fostering organizational green culture is critical for advancing green product innovation [55]. Based on this discussion, we propose the following hypotheses:

H8. In a high-organizational green culture, the relationship between environmental ethics and green process innovation is stronger for manufacturing firms with a high level of organizational green culture.

H9. In low-organizational green culture, the relationship between environmental ethics and sustainable performance diminishes for manufacturing firms with a low level of organizational green culture.

H10. In a high-organizational green culture, the relationship between environmental ethics and green product innovation is stronger for manufacturing firms with a high level of organizational green culture.

2.11. Conceptual Framework

This study contributes to the sustainability literature by addressing critical gaps in the understanding of the mechanisms and conditions under which environmental ethics influences sustainable performance, specifically through green innovation and organizational green culture. While previous research has established links between green innovation and performance [4] and has emphasized the role of organizational green culture in fostering sustainability [3,5], this study uniquely examines organizational green culture as a moderating factor in the environmental ethics–sustainable performance relationship. This approach provides insights into how organizational green culture strengthens or weakens the influence of environmental ethics on both green product and process innovation, ultimately influencing sustainable performance. Additionally, the study advances the understanding by highlighting the mediating roles of green product and process innovation in this relationship, revealing pathways through which environmental ethics contribute to sustainability outcomes. By exploring these complex interactions, this research offers a comprehensive view that not only enriches theoretical perspectives on green innovation and ethics but also provides practical implications for firms seeking to enhance sustainability through strategic cultural and ethical alignments. The conceptual model of this study is shown in Figure 1.



Figure 1. Research model.

3. Methods

3.1. Data Collection and Sampling

This study examines the relationship between environmental ethics and sustainable performance, focusing on organizational green culture within Turkey's manufacturing sector. In 2023, Turkey's manufacturing sector played a significant economic role, accounting for approximately 22.2% of GDP and 20.7% of employment, highlighting its importance to the national economy [56,57]. However, the sector faces challenges including limited financial resources, low production capacity, inflation, and limited technological innovation, which often lead firms to prioritize economic growth over environmental practices [58,59]. Addressing these environmental impacts is essential for emerging economies such as Turkey, which stand to benefit from adopting green practices [60,61].

Data collection followed a cross-sectional design, using the TOBB (The Union of Chambers of Commerce, Industry, Maritime Trade, and Commodity Exchanges of Turkey) database [62] to select a representative sample of 880 manufacturing firms. The initial sample included 1000 firms. Firms with discontinued operations or inaccurate contact information were excluded, leaving 880 firms. Managers from different levels, such as COOs, CFOs, CEOs, HR managers, and production and operations managers, were targeted to gain a comprehensive perspective on green culture and sustainable performance, with each firm represented by a single informant knowledgeable about its environmental practices.

Using a single respondent per firm is a common approach; however, it presents limitations related to social desirability bias, particularly in self-reported assessments of sustainability. Research indicates that respondents often provide answers they believe are expected or socially acceptable, potentially overestimating their sustainable performance [10]. This bias can arise because people and firms tend to report sustainability efforts more favorably than they may actually be, especially on sensitive topics such as environmental responsibility. Additionally, an individual's perception of sustainability behaviors may differ from actual practices, meaning that responses may not fully capture a firm's true activities [63]. Consequently, our study design was unable to fully control for this bias,

Green Innovation

and future studies should consider collecting responses from multiple managerial levels to mitigate its effects.

Incorporating multiple respondents could provide a more balanced view, as relying on a single respondent, even a senior executive, does not guarantee an accurate reflection of all strategic components in an organization [64]. Neiroukh et al. [65] argued that strategic decision-making processes vary by managerial level, with each level providing distinct insights into a firm's strengths, weaknesses, and environmental uncertainties. Therefore, including managers from different levels can offer a fuller understanding of the strategy development process and reduce the single-respondent desirability bias [66]. Multiple participants from the same firm would also enhance scale validation [67] and provide a more accurate view of how strategic processes are perceived across organizational levels.

Data collection was conducted from June to September 2023, with two follow-up reminders, yielding 248 responses, of which 236 were valid, resulting in a 26.8% response rate. To address potential non-response bias, we compared early and late responses and found no significant differences. Additionally, we verified that key characteristics (e.g., the number of employees) were consistent between respondents and a sample of nonrespondents, reinforcing the robustness of our findings [68].

The demographic profile of the respondents (see Table 1) reflected diversity across firm size and industry: 65.7% were from medium-sized firms, 29.2% from large firms, and 5.1% from small firms. Industry representation included food and beverage processing (33.1%), IT and technology (28.4%), textiles and plastics (16.5%), wood and chemicals (14.8%), and pharmaceuticals (7.2%). Furthermore, 55.5% of respondents were middle-level managers, providing valuable insights aligned with the study's focus on sustainable performance.

Variables	Options	Frequency	Percentage (%)
Condon	Male	128	54.2%
Gender	Female	108	45.8%
	Less than 30 years old	2	0.8%
	30–39	61	25.8%
Age	40–49	77	32.6%
	50–59	42	17.8%
	60 years and above	54	22.9%
	Lower-level managers	18	7.6%
Respondent's position	Middle-level manager	131	55.5%
	Top-level managers	87	36.9%
	Pharmaceuticals and medical	17	7 2%
	equipment	17	7.270
	Textile and plastics	39	16.5%
	manufacturing	57	10.5 %
Industry sector	Food and beverage	78	33.1%
industry sector	processing	70	55.170
	Wood and chemical	35	14.8%
	manufacturing	00	11.0 /0
	IT and technological	67	28.4%
	equipment manufacturing	07	20.170
	Small-sized firms	69	29.2%
Number of employees	Medium-sized firms	155	65.7%
	Large-sized firms	12	5.1%
Т	Total	236	100%

Table 1. Respondents' demographic profile.

3.2. Measures

The data were collected using a structured questionnaire. To ensure content validity, the questionnaire was urbanized and back translated from English to Turkish. This process involved scrutiny and refinement of inputs from industrial experts and practitioners. Additionally, a pilot test was conducted among ten manufacturing firms engaged in environmental practices, and feedback from this pilot study was used to make necessary adjustments to enhance the questionnaire's quality and reliability.

The items used to measure the study constructs were derived from previously validated scales (Table 2) and rated on a 7-point Likert scale ranging from 1 (strongly disagree) to 7 (strongly agree). Environmental ethics, the main independent variable, was measured using a 4-item scale adapted from Aftab et al. [10]. This scale effectively captured ethical considerations related to environmental practices within organizations.

The mediating variables, which included green innovations in both the process and product dimensions, were measured using an 8-item scale and each dimension was measured with four items developed by Chen et al. [49] and Aftab et al. [69]. These items assessed the extent of innovative practices aimed at improving the environmental sustainability of manufacturing processes.

The moderating effect of organizational green culture was measured using a 6-item scale developed by Fraj et al. [70] and Wang [71]. This scale evaluated the degree to which companies adopted and institutionalized environmental values through policies, mission statements, and communication programs for employees, thereby reflecting the depth of their green culture [72].

Constructs	Indicators	Sources					
	EE1: My company has specific policies for environmental						
	protection.						
	EE2: My company has a budget for environmental protection.						
Environmental Ethics (EE)	EE3: My company integrates environmental programs,						
	strategies, or objectives into marketing campaigns.						
	EE4: My company integrates environmental programs,						
	strategies, or objectives into its culture.						
	GPD1: The firm chooses the materials of the product that						
	produce the least amount of pollution for conducting the						
	product development or design.	[49,69]					
	GPD2: The firm chooses the materials of the product that						
Croop Product Inposetion	consume the least amount of energy and resources for						
(CPD)	conducting the product development or design.						
(GFD)	GPD3: The firm uses the fewest number of materials to comprise						
	the product for conducting the product development or design.						
	GPD4: The firm would circumspectly deliberate whether the						
	product is easy to recycle, reuse, and decompose for conducting						
	the product development or design.						
	GPI1: The manufacturing process of the firm effectively reduces						
	the emission of hazardous substances or waste.						
Green Process Innovation	GPI2: The manufacturing process of the firm recycles waste and						
	emissions that allow them to be treated and reused.	[40,60]					
(GPI)	GPI3: The manufacturing process of the firm reduces the	[49,69]					
	consumption of water, electricity, coal, or oil.						
	GPI4: The manufacturing process of the firm reduces the use of						
	raw materials.						

 Table 2. Measurement items.

	OGC1: Our firm makes a concerted effort to make every employee understand the importance of environmental preservation			
	OCC2: Our firm has a clear policy statement urging			
	environmental awareness in every area			
	OCC3: Environmental preservation is a high-priority activity in			
Organizational Green	our firm	[70 71]	[70 71]	
Culture (OGC)	OCC4: Preserving the environment is a central corporate value	[/0,/1]		
	in our firm			
	OCC5: Our firm links anyironmental objectives with our other			
	corporate goals			
	porate goals.			
	ogco. Our mini develops products and processes that minimize			
Sustainable Parformance (SI				
Sustainable renormance (SC	CUD1. Deduction of air omissions			
	SUP1: Reduction of all emissions.			
Environmental performance	SUP2: Reduction of effluent and solid wastes.			
(ENP)	SUP3: Reduced consumption of hazardous/harmful/toxic			
	materials.			
	SUP4: Improved social responsibility towards human rights.			
	SUP5: Improved social responsibility towards the safety of	[58,73]		
Social performance (SOP)	employees.			
	SUP6: Improved personal desires of employees to do what is			
	right.			
F : (SUP7: Reduced in cost of materials purchasing.			
Economic performance	SUP8: Reduced in cost for energy consumption.			
(ECP)	SUP9: Reduced fee for waste treatment.			

Sustainable performance, the focal outcome of this study, was assessed using a 9-item scale by Abuzawida et al. [58] and Yavuz et al. [73], which encompasses three dimensions: environmental performance, social performance, and economic performance [10,74,75]. This comprehensive scale provided a nuanced understanding of how sustainability initiatives impact various aspects of organizational performance.

3.3. Common Method Bias

To ensure the reliability of the findings, the study addressed common method bias, which can occur when data are collected from a single source and potentially skew the results if unaddressed [76]. The researchers applied Harman's one-factor test, a widely recognized method for detecting common method bias, which indicates its presence if a single factor accounts for 50% or more of the variance [77]. In this study, the first factor explained only 34.67% of the variance, suggesting a minimal bias. Additionally, the researchers conducted a full collinearity test, as recommended by Kock [78], with variance inflation factors (VIFs) \leq 5 (see Table 3), confirming that multicollinearity and standard method bias were not significant issues [79]. These steps ensured that the study's conclusions about the relationship between environmental ethics, organizational green culture, and sustainable performance are robust and credible.

Construct/Indicators	Outer Loadings	Means	Standard Deviation	Standard VIFs Deviation		CR	AVE
Environmental Ethics (EE)					0.856	0.903	0.698
EE1	0.859	5.08	1.178	2.112			
EE2	0.834	5.22	1.312	2.055			
EE3	0.831	5.26	1.284	1.952			
EE4	0.819	5.40	1.179	1.793			
Green Product Inne	ovation (GPD)				0.846	0.896	0.683
GPD1	0.781	5.53	1.089	1.526			
GPD2	0.816	5.41	1.150	2.070			
GPD3	0.848	5.44	1.127	2.546			
GPD4	0.860	5.36	1.124	2.279			
Green Process Inn	ovation (GPI)				0.813	0.871	0.628
GPI1	0.752	5.28	1.236	1.828			
GPI2	0.804	5.54	1.299	2.300			
GPI3	0.817	5.81	1.198	2.097			
GPI4	0.795	5.47	1.193	1.322			
Organizational Green	n Culture (OGC)				0.770	0.832	0.555
OGC1	0.794	5.31	1.285	2.258			
OGC2	0.700	5.35	1.275	2.099			
OGC3	0.691	5.20	1.269	1.735			
OGC4	0.632	5.18	1.564	2.751			
OGC5	0.645	5.23	1.525	2.955			
OGC6	0.561	5.22	1.588	2.447			
Sustainable Perfor	rmance (SUP)				0.924	0.937	0.623
Environmental perfe	ormance (ENP)				0.868	0.919	0.791
SUP1	0.870	4.83	1.393	2.166			
SUP2	0.911	4.57	1.524	2.587			
SUP3	0.887	4.81	1.523	2.199			
Social performa	ance (SOP)				0.798	0.881	0.713
SUP4	0.865	5.07	1.223	1.940			
SUP5	0.861	5.03	1.417	1.907			
SUP6	0.806	4.97	1.396	1.448			
Economic perform	mance (ECP)				0.852	0.910	0.772
SUP7	0.877	4.94	1.337	2.129			
SUP8	0.866	4.78	1.301	2.429			
SUP9	0.892	4.87	1.334	2.206			

reliability and convergent validity.
reliability and convergent validity

Note(s): Variance inflation factors (VIFs), Cronbach's alpha (CA), composite reliability (CR), average variance extracted (AVE).

4. Analysis and Results

The analysis of this study was conducted using partial least squares structural equation modeling (PLS-SEM), a robust statistical technique well regarded for its ability to handle complex reflective models without requiring strict distributional assumptions [79]. PLS-SEM is particularly valuable in the fields of operations and green management as it efficiently handles data noise and tests causal relationships between constructs [80]. This method was chosen because of its high predictive accuracy and effectiveness in evaluating both the measurement model's reliability and hypothesized relationships of the structural model [81]. Through PLS-SEM, this study rigorously assessed the reliability and validity of the instruments used and tested the proposed hypotheses, providing a comprehensive understanding of how environmental ethics, mediated by organizational green culture, is associated with sustainable performance.

4.1. Reliability and Validity of the Measurement Model

To ensure the robustness of the measurement model, its reliability and validity were meticulously assessed. Construct reliability was determined using Cronbach's alpha (CA) and composite reliability (CR), with both metrics surpassing the recommended threshold of 0.70 (Table 3), indicating strong internal consistency [79]. Convergent validity was confirmed through factor loadings and average variance extracted (AVE); all factor loadings were above 0.50, and AVE values exceeded the minimum requirement of 0.50, ensuring that the constructs were well measured [82]. Discriminant validity was established using the Heterotrait-Monotrait Ratio (HTMT), with all values below the 0.9 threshold (see Table 4), affirming that the constructs were distinct from one another [83]. These steps collectively confirmed the reliability and validity of the measurement model, providing a solid foundation for subsequent hypothesis testing.

EE	GPD	GPI	OGC	SUP
0				
0.470	0			
0.394	0.826	0		
0.523	0.253	0.217	0	
0.627	0.415	0.411	0.638	0
	EE 0 0.470 0.394 0.523 0.627	EE GPD 0 0 0.470 0 0.394 0.826 0.523 0.253 0.627 0.415	EE GPD GPI 0 0 0 0.470 0 0 0.394 0.826 0 0.523 0.253 0.217 0.627 0.415 0.411	EE GPD GPI OGC 0

Table 4. Discriminant validity (HTMT).

4.2. Assessment of the Structural Model

The structural model was meticulously assessed to ensure the reliability and validity of the hypothesized relationships. Multicollinearity was first examined and found to be non-problematic, with all VIFs \leq 5, indicating no significant collinearity issues [78,81]. This study employed a bootstrapping procedure with 5000 resamples to determine the significance of the model's path coefficients [84].

As shown in Table 5 of Figure 2, the results demonstrated that environmental ethics was significantly and positively associated with sustainable performance ($\beta = 0.240$, t = 4.537, p = 0.000), supporting H1. Additionally, environmental ethics significantly enhanced green process innovation ($\beta = 0.399$, t = 6.986, p = 0.000), thus supporting H2. In turn, green process innovation was significantly related to sustainable performance ($\beta = 0.197$, t = 3.169, p = 0.002), supporting H3. Furthermore, environmental ethics significantly promoted green product innovation ($\beta = 0.437$, t = 6.431, p = 0.000), thus supporting H4. However, green product innovation was not significantly associated with sustainable performance ($\beta = 0.049$, t = 0.732, p = 0.464), thus providing no support for H5. These results underscored the vital role of environmental ethics and process innovation in driving sustainable performance; however, the direct impact of product innovation may require further exploration.



Figure 2. Structural model estimation. Note(s): Environmental Ethics (EE), Green Product Innovation (GPD), Green Process Innovation (GPI), Organizational Green Culture (OGC), Sustainable Performance (SUP).

Table 5. Direct relationship results.

Hypothesis	Path coefficient	t-Value	<i>p</i> -Value	Result
H1: $EE \rightarrow SUP$	0.240	4.537	0.000	Supported
H2: $EE \rightarrow GPI$	0.399	6.986	0.000	Supported
H3: GPI \rightarrow SUP	0.197	3.169	0.002	Supported
H4: $EE \rightarrow GPD$	0.437	6.431	0.000	Supported
H5: GPD \rightarrow SUP	0.049	0.732	0.464	Not supported

Note(s): Environmental Ethics (EE), Green Product Innovation (GPD), Green Process Innovation (GPI), Sustainable Performance (SUP).

4.3. Mediation Analysis of Green Innovation

Mediation analysis aimed to determine whether green process innovation and green product innovation mediate the relationship between environmental ethics and sustainable performance. Following the method validated by [85,86], this study compared the path coefficients and the significance of the direct and indirect effects. The results presented in Table 6 show that green process innovation provided a partial mediation effect between environmental ethics and sustainable performance ($\beta = 0.079$, t = 2.820, *p* = 0.005), supporting H6. This indicated that environmental ethics directly and indirectly enhanced sustainable performance by fostering green process innovation. However, green product innovation did not mediate this relationship ($\beta = 0.021$, t = 0.719, *p* = 0.472), indicating that H7 was not supported. This suggested that while environmental ethics promoted green product innovation, the latter was not significantly related to sustainable performance within the scope of this study, potentially because of the longer timeframes needed for product innovations to impact sustainability metrics.

Table 6. Indirect and interaction relationship results.

Path	Path Coefficient	<i>t</i> -Value	CI 2.5%	CI 97.5%	<i>p</i> -Value	Result
H6: $EE \rightarrow GPI \rightarrow SUP$	0.079	2.820	0.028	0.137	0.005	Supported
$H7: EE \rightarrow GPD \rightarrow SUP$	0.021	0.719	-0.038	0.079	0.472	Not supported

H8: EE × OGC \rightarrow GPI	0.168	2.412	0.025	0.297	0.016	Supported
H9: EE × OGC \rightarrow SUP	-0.131	3.713	-0.197	-0.057	0.000	Supported
H10: EE × OGC \rightarrow GPD	0.132	2.198	0.005	0.241	0.028	Supported
	Nota(a): Environmental	Ethics (EE)	Croop Product	Innovation (CPD	Croon	Process Innovation

Note(s): Environmental Ethics (EE), Green Product Innovation (GPD), Green Process Innovatior (GPI), Organizational Green Culture (OGC), Sustainable Performance (SUP).

4.4. Moderation Role Results of Organizational Green Culture

This study examined the moderating effects of organizational green culture on the relationships among environmental ethics, green innovation, and sustainable performance. Using the product indicator method [87–89], the PLS-SEM results in Table 6 revealed that organizational green culture positively moderated the relationship between environmental ethics and green process innovation (β = 0.168, t = 2.412, *p* = 0.016), supporting H8. This finding, illustrated in Figure 3, indicated that the positive relationship between environmental ethics and green process innovation was stronger for firms with a high level of green culture. This suggested that a strong green culture within an organization enhanced the effectiveness of ethical environmental practices, leading to greater innovation in green processes.



Figure 3. The interaction role of organizational green culture (OGC) in the relationship between environmental ethics (EE) and green process innovation (GPI).

Conversely, organizational green culture negatively moderated the link between environmental ethics and sustainable performance ($\beta = -0.131$, t = 3.713, *p* = 0.000), supporting H9. As depicted in Figure 4, this suggested that, in firms with a low level of green culture, the positive relationship between environmental ethics and sustainable performance was diminished. This counterintuitive finding might reflect the challenges faced by firms with a weak green culture in translating ethical practices into tangible sustainability outcomes, possibly because of a lack of supportive infrastructure or employee engagement.



Figure 4. The interaction role of organizational green culture (OGC) in the relationship between environmental ethics (EE) and sustainable performance (SUP).

Additionally, organizational green culture positively moderated the relationship between environmental ethics and green product innovation (β = 0.132, t = 2.198, *p* = 0.028), supporting H10. Figure 5 illustrates that this relationship was stronger in firms with a high level of organizational green culture. This underscored the role of an ingrained green culture in fostering product innovation that aligns with environmental ethics, potentially leading to the development of eco-friendly products and sustainable business practices. Collectively, these findings underscored the critical role of organizational green culture in enhancing the effectiveness of environmental ethics in fostering green innovation and achieving sustainable performance in the manufacturing sector.



Figure 5. The interaction role of organizational green culture (OGC) in the relationship between environmental ethics (EE) and green product innovation (GPD).

5. Discussion

This study investigated the relationship between environmental ethics and sustainable performance, emphasizing the mediating roles of green process and product innovations, as well as the moderating role of organizational green culture within the Turkish manufacturing industry. The results contribute to the increasing body of literature on environmental ethics and sustainability by demonstrating how ethical business practices can enhance sustainability outcomes.

The results confirm a positive correlation between environmental ethics and sustainable performance, indicating that companies prioritizing ethical practices tend to achieve better sustainability results. This aligns with recent studies highlighting that organizational ethics, particularly environmental responsibility, significantly drives sustainable development [10]. This study supports the resource-based view framework by positioning ethical practices as internal resources that provide firms with a competitive advantage, subsequently leading to improved long-term performance. In this context, environmental ethics serves as a valuable asset, enabling firms to mitigate environmental risks, comply with regulations, and satisfy stakeholder expectations for enhanced sustainable performance [90].

The mediation investigation indicated that green process innovation significantly mediates the link between environmental ethics and sustainable performance. This finding reinforces previous research indicating that organizations adopting green process innovations, such as waste reduction, energy efficiency, and eco-friendly production methods, are better equipped to meet their sustainability goals. By embedding environmental ethics into their operational strategies, firms not only improve sustainability performance but also enhance their overall efficiency and competitiveness [90]. The positive mediating role of green process innovation suggests that companies with strong ethical foundations are more likely to support process innovation, resulting in notable improvements in both environmental and economic performance [63]. Conversely, green product innovation did not mediate the relationship between environmental ethics and sustainable performance, contradicting recent literature that emphasizes the growing importance of green product innovation in sustainability efforts [91]. This discrepancy may be attributed to the longer timeframes required for the benefits of green product innovation to materialize compared to green process innovation. Developing green products often necessitates substantial investments in research and development (R&D), longer product life cycles, and extended market adaptation periods, potentially delaying the realization of sustainability benefits [27]. Although firms may invest in green product innovation, the time required for these innovations to translate into measurable sustainability outcomes may extend beyond the scope of this study.

A significant moderating role of organizational green culture was identified, whereby firms with a strong green culture amplified the positive association of environmental ethics with both green process and product innovations. This finding supports recent research that highlights the critical role of organizational culture in fostering innovation and advancing sustainability [55]. Companies that embed green values within their organizational culture are more likely to create an environment conducive to promoting and supporting innovations aligned with ethical practices. Studies have demonstrated that a robust organizational green culture not only enhances environmental performance but also strengthens a firm's competitive advantage through innovation [92]. Notably, organizations with a weak green culture exhibited diminished relationships between environmental ethics and sustainable performance. This suggests that, without a supportive organizational culture, firms may struggle to translate their ethical commitments into tangible sustainability outcomes. The findings introduce a new perspective in the ongoing debate on the role of culture in sustainability, positing that ethical behavior alone is insufficient to achieve sustainability goals without a strong supporting green culture. In companies lacking a green culture, even well-intentioned ethical practices may fail to yield the desired sustainability results, potentially owing to inadequate engagement, coordination, or strategic alignment [2].

This study empirically confirms the integration of environmental ethics with innovative practices to achieve sustainable performance. While previous research has often explored these concepts in isolation, this study demonstrates that ethical practices, innovation, and culture are interconnected and collaboratively drive sustainability [10]. Specifically, the mediating role of green process innovation underscores operational improvements that reduce environmental impacts while increasing organizational efficiency. Concurrently, the moderating role of organizational green culture indicates that firms with strong environmental values are better positioned to convert their ethical commitments into advantages in innovation and sustainability. To capitalize on these findings, managers should foster a robust green organizational culture while implementing ethical practices. By embedding green values within the organizational culture, firms can cultivate an environment that inspires and nurtures innovation, ultimately leading to enhanced sustainable performance. Furthermore, this research highlights the relevance of green process innovation as an efficient pathway to achieving sustainability, particularly in industries where product innovation may take longer to realize [91]. This study elaborates on the complex interactions among environmental ethics, green innovation, and sustainable performance, emphasizing the critical roles of ethical practices and organizational green culture in driving sustainability while illustrating the distinct contributions of process and product innovations within this dynamic.

6. Conclusions

6.1. Theoretical Implications

This study makes important theoretical contributions to the understanding of sustainable performance and green innovation, particularly in Turkey's manufacturing sector. By integrating environmental ethics, organizational green culture, and both green process and product innovations into a comprehensive framework, this study links ethical business practices to sustainable outcomes based on stakeholder theory and the resourcebased view. This framework enhances the existing literature by highlighting the moderating role of organizational green culture and mediating role of green innovation.

The first contribution of this research is its extension of the existing knowledge regarding the link between environmental ethics and sustainable performance. Studies have primarily focused on ecological supply chains or technological innovation, emphasizing the ethical foundations necessary for firms to achieve sustainable performance [1]. This demonstrates that environmental ethics is positively associated with both green process and product innovations, thereby addressing a significant gap in the literature on ethical imperatives for sustainable innovation. This finding challenges the traditional economic perspective that profit maximization impedes sustainability, suggesting that ethical business frameworks can drive competitive advantages [4].

Additionally, this study distinguishes green process innovation and green product innovation as mediating factors in the relationship between environmental ethics and sustainable performance. Although previous research has examined the relationship between green innovation and firm performance, few studies have investigated the separate roles of process and product innovations in promoting sustainable performance [10]. This study provides empirical evidence that while both forms of innovation mediate this relationship, green process innovation has a stronger and more immediate impact on sustainable performance than green product innovation. This finding aligns with earlier research indicating that operational improvements are crucial for achieving sustainability [91].

Furthermore, this study contributes to the theoretical discourse on organizational green culture by identifying its significant moderating role in the relationship between environmental ethics, green innovation, and sustainable performance. Previous studies have suggested that corporate culture fosters sustainability; however, this research offers

robust empirical support for this claim [55]. This finding suggests that firms with a strong green culture derive greater benefits from ethical leadership. This underscores the notion that ethical behavior alone may not suffice to achieve high levels of sustainability without enabling organizational culture.

In terms of stakeholder theory, this research provides insights into the manufacturing sector in Turkey, an emerging economy facing unique challenges, such as resource constraints and environmental degradation [10]. This illustrates how stakeholder power encompassing consumers, regulators, and non-governmental organizations drives firms to adopt green innovations and align ethical practices with sustainability goals. This research broadens the application of stakeholder theory, extending its discourse from a Western context to new insights relevant to emerging markets [2]. Finally, this study enriches the resource-based view by positing that environmental ethics is a vital internal resource for firms to develop distinctive capabilities, especially in the realm of green innovation [93]. These findings support earlier studies asserting that green capabilities can confer sustained competitive advantages. By integrating the resource-based view with stakeholder theory, this study creates a holistic framework that elucidates how firms can overcome sustainability challenges while maintaining competitive market positions [24,94].

This study connects environmental ethics, green innovation, and sustainable performance, highlighting the mediating roles of both process and product innovations and the moderating role of organizational green culture. It offers new insights into how ethical practices and organizational culture can facilitate sustainability, particularly in emerging market contexts.

6.2. Practical and Managerial Implications

This study offers significant practical and managerial implications for firms seeking to enhance their sustainable performance by integrating environmental ethics, green innovation, and organizational culture. The findings emphasize the importance of focusing on green process innovation and fostering green organizational culture as essential drivers of sustainability, which is particularly relevant for managers in environmentally conscious industries.

This study highlights the necessity of embedding environmental ethics in core business strategies. Prior research indicates that prioritizing environmental ethics can enhance corporate reputation, ensure regulatory compliance, and align with the increasing consumer demand for green products [2,10]. From a managerial perspective, environmental ethics should be viewed not only as a moral obligation but also as a strategic advantage that strengthens stakeholder relationships and reduces operational risks [49]. Firms that prioritize environmental ethics can better anticipate regulatory challenges and meet the needs of eco-conscious consumers, ultimately improving their economic performance [90].

Green process innovation serves as a crucial mechanism through which environmental ethics translates into sustainable performance [27,91]. Managers are encouraged to prioritize the adoption of greener process innovations, such as energy-efficient manufacturing and waste reduction initiatives, because these innovations can decrease environmental footprints and operational costs, while enhancing efficiency. Investments in green process innovation are likely to yield long-term sustainability benefits, especially in resource-intensive industries, where substantial cost savings can be achieved [10]. While green product innovation offers significant advantages, these benefits may take longer to materialize compared to process innovation. Recent research indicates that substantial investment in R&D and extended market adaptation periods are required for successful green product development. Consequently, managers should consider green product innovation from a long-term perspective [27]. Although immediate benefits may be delayed, investing in green product innovation positions firms as leaders in sustainability, providing a competitive edge as the demand for eco-friendly products continues to grow [2].

The findings also illustrate the significant moderating role of organizational green culture on the link between environmental ethics and sustainable performance [92].

Industries with a robust organizational green culture are enhanced in order to effectively implement green innovations; as such, the culture fosters an atmosphere that is favorable to sustainable performance and innovation [55]. This highlights the importance of leader-ship commitment to sustainability, employee engagement in environmental initiatives, and the integration of environmental values into an organization's mission and vision. Without a supportive green culture, even the best-intentioned ethical practices may fall short of driving meaningful sustainability outcomes. Managers should focus on cultivating this culture through training programs, sustainability initiatives, and encouraging employee participation in green practices [2].

In organizations lacking a green culture, the influence of environmental ethics on sustainable performance may be limited. To address this, managers should prioritize cultural transformation by promoting sustainability initiatives at all levels and creating a framework that supports ethical practices [92]. This can be achieved through active leadership, sustainability-focused performance metrics, and clear communication regarding environmental goals. This research aligns with the United Nations Sustainable Development Goals (SDGs), particularly Goals 12 (Responsible Consumption and Production) and 13 (Climate Action). By prioritizing environmental ethics and green innovation, firms can reduce their environmental impact and support global objectives for responsible consumption and climate change. Managers can leverage these findings to enhance organizational sustainability, while contributing to broader societal goals for a sustainable future.

6.3. Limitations and Future Studies

This study contributes to the understanding of how environmental ethics and green innovation are associated with sustainable performance, moderated by organizational green culture. However, there are several limitations that open avenues for future research. First, the cross-sectional design restricts the ability to infer causality, as it captures relationships at a single point in time [77]. Future studies could adopt longitudinal designs to better understand how green innovations evolve over time and their sustained impacts on performance. Second, while this study used self-reported survey data from manufacturing firms, this approach may oversimplify the complexity of green innovation and sustainable performance, given the potential for social desirability bias in self-reported data [66]. High-quality secondary datasets, such as the Community Innovation Survey and Eurobarometer flash survey, offer rich modules for green innovations and could provide more robust insights into sustainable-led innovation practices [95–97]. Third, the theoretical model focuses on organizational green culture and green innovation but does not account for critical external factors such as regulatory frameworks, market dynamics, and stakeholder pressures, which are essential drivers of sustainability strategies [27,69]. Future research could integrate these external determinants to provide a more comprehensive view of sustainability. Finally, while this study assesses sustainable performance across environmental, social, and economic dimensions, a deeper exploration of how these dimensions interact with broader SDGs is warranted [58,73]. Advanced analytical methods, such as mixed-method approaches and machine learning, can offer nuanced insights into these complex interrelations and aid in developing innovative strategies for sustainable performance. By addressing these limitations, future research can build on these findings to offer a more holistic and actionable framework for enhancing sustainability in diverse organizational contexts.

Author Contributions: Validation, H.Y.A. and K.I.; Writing—original draft, E.E.; Supervision, A.A.; Project administration, A.A. All authors have read and agreed to the published version of the manuscript.

Funding: This study received no external funding.

Institutional Review Board Statement: The study was conducted in accordance with the Declaration of Helsinki and by the University of Mediterranean Karpasia Institutional Review Board (IRB) (protocol code: 2023-24-Spring-03 and date of approval: 3 May 2023).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Raw data supporting the conclusions of this article will be made available by the authors upon request.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Mostepaniuk, A.; Nasr, E.; Awwad, R.I.; Hamdan, S.; Aljuhmani, H.Y. Managing a Relationship between Corporate Social Responsibility and Sustainability: A Systematic Review. *Sustainability* 2022, 14, 11203.
- 2. Imran, M.; Arshad, I.; Ismail, F. Green organizational culture and organizational performance: The mediating role of green innovation and environmental performance. *J. Pendidik. IPA Indones.* **2021**, *10*, 515–530.
- 3. Kraus, S.; Rehman, S.U.; García, F.J.S. CSR and environmental performance: The mediating role of environmental strategy and green innovation. *Technol. Forecast. Soc. Change* **2020**, *160*, 120262.
- 4. Singh, S.K.; Del Giudice, M.; Chierici, R.; Graziano, D. Green innovation and environmental performance: The role of green transformational leadership and green human resource management. *Technol. Forecast. Soc. Chang.* **2020**, *150*, 119762.
- Adomako, S.; Amankwah-Amoah, J. Managerial attitude towards the natural environment and environmental sustainability expenditure. J. Clean. Prod. 2021, 326, 129384.
- 6. Guo, H.; Lu, W. The inverse U-shaped relationship between CSR and competitiveness: Evidence from Chinese international construction companies. J. Clean. Prod. 2021, 295, 126374.
- 7. Yin, C.; Ma, H.; Gong, Y.; Chen, Q.; Zhang, Y. Environmental CSR and environmental citizenship behavior: The role of employees' environmental passion and empathy. *J. Clean. Prod.* **2021**, *320*, 128751.
- Rui, Z.; Lu, Y. Stakeholder pressure, corporate environmental ethics, and green innovation. *Asian J. Technol. Innov.* 2021, 29, 70– 86.
- 9. Padilla-Lozano, C.P.; Collazzo, P. Corporate social responsibility, green innovation and competitiveness–causality in manufacturing. *Compet. Rev. Int. Bus. J.* **2022**, *32*, 21–39.
- Aftab, J.; Abid, N.; Sarwar, H.; Veneziani, M. Environmental ethics, green innovation, and sustainable performance: Exploring the role of environmental leadership and environmental strategy. J. Clean. Prod. 2022, 378, 134639.
- Kara, E.; Akbaba, M.; Yakut, E.; Çetinel, M.H.; Pasli, M.M. The mediating effect of green human resources management on the relationship between organizational sustainability and innovative behavior: An application in Turkey. *Sustainability* 2023, 15, 2068.
- 12. Freeman, R.E. Strategic Management: A Stakeholder Approach; Cambridge University Press: Cambridge, UK, 2010.
- 13. Emeagwali, O.L.; Aljuhmani, H.Y. Strategic Management: A Dynamic View; IntechOpen: London, UK, 2019.
- 14. Mdolo, M.P.; Naghavi, N.; Fah, B.C.Y.; Jin, T.C. Corporate sustainability adoption amongst public listed companies in Malaysia. *Turk. Online J. Des. Art Commun.* **2018**, *8*, 509–517.
- 15. Asha'ari, M.J.; Daud, S.; Suki, N.M. Linking sustainable design and social sustainability performance of chemical manufacturing firms: The moderating role of Islamic work ethics. *Sustainability* **2023**, *15*, 5991.
- 16. Alsafadi, Y.; Aljuhmani, H.Y. The Influence of Entrepreneurial Innovations in Building Competitive Advantage: The Mediating Role of Entrepreneurial Thinking. *Kybernetes* **2024**, *53*, 4051–4073. https://doi.org/10.1108/K-11-2022-1540.
- 17. Hart, S.L.; Dowell, G. Invited editorial: A natural-resource-based view of the firm: Fifteen years after. J. Manag. 2011, 37, 1464– 1479.
- 18. Barney, J. Firm resources and sustained competitive advantage. J. Manag. 1991, 17, 99–120.
- 19. Wagner, M.; Van Phu, N.; Azomahou, T.; Wehrmeyer, W. The relationship between the environmental and eco-nomic performance of firms: An empirical analysis of the European paper industry. *CSR Environ. Manag.* **2002**, *9*, 133–146.
- 20. Ployhart, R.E. The psychology of competitive advantage: An adjacent possibility. Ind. Organ. Psychol. 2012, 5, 62-81.
- 21. Zhu, Q.; Geng, Y.; Fujita, T.; Hashimoto, S. Green supply chain management in leading manufacturers: Case studies in Japanese large companies. *Manag. Res. Rev.* 2010, *33*, 380–392.
- 22. Sarkis, J.; González-Torre, P.; Adenso-Díaz, B. Stakeholder pressure and the adoption of environmental practices: The mediating effect of training. *J. Oper. Manag.* 2010, *28*, 163–176.
- 23. Chen, Y.S. The driver of green innovation and green image-Green core competence. J. Bus. Ethics 2008, 81, 531-543.
- 24. Porter, M.E.; van der Linde, C. Green and competitive: Ending the stalemate. *Harv. Bus. Rev.* **1995**, 73, 120–134.
- 25. Zameer, H.; Wang, Y.; Yasmeen, H.; Mubarak, S. Green innovation as a mediator in the impact of business analytics and environmental orientation on green competitive advantage. *Manag. Decis.* **2022**, *60*, 488–507.
- Guo, Y.; Wang, L.; Yang, Q. Do corporate environmental ethics influence firms' green practices? The mediating role of green innovation and the moderating role of personal ties. *J. Clean. Prod.* 2020, 266, 122054.
- Alshammari, T.A.; Alshammary, K.M.; Alshammari, F.M. Green Product and Process Innovation, Corporate environmental ethics and Competitive Advantages among Manufacturing Firms in the Kingdom of Saudi Arabia. J. Econ. Behav. Stud. 2022, 14, 51–61.

- Song, W.; Yu, H. Green innovation strategy and green innovation: The roles of green creativity and green organizational identity. *Corp. Soc. Responsib. Environ. Manag.* 2018, 25, 135–150.
- Shehzad, K.; Liu, X.; Sarfraz, M.; Zulfiqar, M. Signifying the imperative nexus between climate change and information and communication technology development: A case from Pakistan. *Environ. Sci. Pollut. Res.* 2020, 27, 30502–30517.
- Kuzma, E.; Padilha, L.S.; Sehnem, S.; Julkovski, D.J.; Roman, D.J. The relationship between innovation and sustainability: A meta-analytic study. J. Clean. Prod. 2020, 259, 120745.
- Sarfraz, M.; Ivascu, L.; Cioca, L.-I. Environmental Regulations and CO₂ Mitigation for Sustainability: Panel Data Analysis (PMG, CCEMG) for BRICS Nations. *Sustainability* 2021, 14, 72.
- 32. Luo, Y.; Salman, M.; Lu, Z. Heterogeneous impacts of environmental regulations and foreign direct investment on green innovation across different regions in China. *Sci. Total Environ.* **2021**, *759*, 143744.
- Sarfraz, M.; Ivascu, L.; Abdullah, M.I.; Ozturk, I.; Tariq, J. Exploring a pathway to sustainable performance in manufacturing firms: The interplay between innovation capabilities, green process, product innovations and digital leadership. *Sustainability* 2022, 14, 5945.
- Sellitto, M.A.; Camfield, C.G.; Buzuku, S. Green innovation and competitive advantages in a furniture industrial cluster: A survey and structural model. *Sustain. Prod. Consum.* 2020, 23, 94–104.
- 35. Awan, U.; Nauman, S.; Sroufe, R. Exploring the effect of buyer engagement on green product innovation: Empirical evidence from manufacturers. *Bus. Strategy Environ.* **2021**, *30*, 463–477.
- Appiah, L.O.; Essuman, D. How do firms develop and financially benefit from green product innovation in a developing country? Roles of innovation orientation and green marketing innovation. *Bus. Strategy Environ.* 2024, 33, 7241–7252.
- Song, W.; Wang, G.; Ma, X. Environmental innovation practices and green product innovation performance: A perspective from organizational climate. *Sustain. Dev.* 2020, 28, 224–234.
- Ma, Y.; Yin, Q.; Pan, Y.; Cui, W.; Xin, B.; Rao, Z. green product innovation and firm performance: Assessing the moderating effect of novel-ty-centered and efficiency-centered business model design. *Sustainability* 2018, 10, 1843.
- Muangmee, C.; Dacko-Pikiewicz, Z.; Meekaewkunchorn, N.; Kassakorn, N.; Khalid, B. Green Entrepreneurial Orientation and green innovation in Small and Medium-Sized Enterprises (SMEs). Soc. Sci. 2021, 10, 136.
- 40. Elzek, Y.; Gaafar, H.; Abdelsamie, H. The Impact of green innovation on Sustainability Performance in Travel Agencies and Hotels: The Moderating Role of Environmental Commitment. *Int. J. Hosp. Tour. Syst.* **2021**, *14*, 15–24.
- Karabulut, T.; Hatipoğlu, H.N. The Effects of green product innovation and green process innovation on Company Performance. *Int. J. Commer. Financ.* 2020, *6*, 181–193.
- 42. Zhang, Y.; Wei, F. SMEs' charismatic leadership, product life cycle, environmental performance, and financial performance: A mediated moderation model. J. Clean. Prod. 2021, 306, 127147.
- Wang, M.; Li, Y.; Li, J.; Wang, Z. Green process innovation, green product innovation and its economic performance improvement paths: A survey and structural model. J. Environ. Manag. 2021, 297, 113282.
- Tang, M.; Walsh, G.; Lerner, D.; Fitza, M.A.; Li, Q. Green innovation, Managerial Concern and Firm Performance: An Empirical Study. Bus. Strateg. Environ. 2018, 27, 9–51.
- Baah, C.; Opoku-Agyeman, D.; Acquah, I.S.K.; Agyabeng-Mensah, Y.; Afum, E.; Faibil, D.; Abdoulaye, F.A.M. Examining the correlations between stakeholder pressures, green production practices, firm reputation, environmental and financial performance: Evidence from manufacturing SMEs. *Sustain. Prod. Consum.* 2021, 27, 100–114.
- 46. Putri, W.H.; Sari, N.Y. Eco-efficiency and eco-innovation: Strategy to improve sustainable environmental performance. *IOP Conf. Ser. Earth Environ. Sci.* 2019, 245, 012049.
- 47. Khairani, S.; Susetyo, D.; Yusnaini, E.; Yusrianti, H. The Effect of green process innovation on Corporate Sustainability and Environmental Performance as a Mediation Variable. In Proceedings of the 7th Sriwijaya Economics, Accounting, and Business Conference (SEABC 2021), Online, 16–17 November 2021; Atlantis Press: Amsterdam, The Netherlands, 2022; pp. 147–154.
- 48. Chen, Y.S.; Chang, C.H. The determinants of green product development performance: Green dynamic capabilities, green transformational leadership, and green creativity. *J. Bus. Ethics* **2013**, *116*, 107–119.
- 49. Chen, Y.S.; Lai, S.B.; Wen, C.T. The influence of green innovation performance on corporate advantage in Taiwan. *J. Bus. Ethics* **2006**, *67*, 331–339.
- 50. Chen, C. Design for the environment: A quality-based model for green product development. Manag. Sci. 2001, 47, 250–263.
- 51. Chen, Y.S.; Chang, T.W.; Lin, C.Y.; Lai, P.Y.; Wang, K.H. The influence of proactive green innovation and reactive green innovation on green product development performance: The mediation role of green creativity. *Sustainability* **2016**, *8*, 966.
- 52. Chen, W.J. Innovation in Hotel Services: Culture and Personality. Int. J. Hosp. Manag. 2011, 30, 64–72.
- 53. Gurlek, M.; Koseoglu, M.A. Green innovation research in the Field of Hospitality and Tourism: The Construct, Antecedents, Consequences, and Future Outlook. *Serv. Ind. J.* **2021**, *41*, 734–766.
- 54. Tajeddini, K.; Trueman, M. Managing Swiss Hospitality: How Cultural Antecedents of Innovation and Customer-Oriented Value Systems Can Influence Performance in the Hotel Industry. *Int. J. Hosp. Manag.* **2012**, *31*, 1119–1129.
- 55. Şengüllendi, M.F.; Bilgetürk, M.; AfacanFındıklı, M. Ethical leadership and green innovation: The mediating role of green organizational culture. *J. Environ. Plan. Manag.* **2024**, *67*, 1702–1723.
- International Trade Administration. Turkey—Advanced Manufacturing. 6 January 2024. Available online: https://www.trade.gov/country-commercial-guides/turkey-advanced-manufacturing (accessed on 25 September 2024).

- TÜİK Data Portal. TÜİK Veri Portalı. 2023. Available online: https://data.tuik.gov.tr/Kategori/GetKategori?p=Employment,-Unemployment-and-Wages-108 (accessed on 25 September 2024).
- Abuzawida, S.S.; Alzubi, A.B.; Iyiola, K. Sustainable Supply Chain Practices: An Empirical Investigation from the Manufacturing Industry. *Sustainability* 2023, 15, 19. https://doi.org/10.3390/su151914395.
- Issa, A.; Khadem, A.; Alzubi, A.; Berberoğlu, A. The Path from green innovation to Supply Chain Resilience: Do Structural and Dynamic Supply Chain Complexity Matter? Sustainability 2024, 16, 99. https://doi.org/10.3390/su16093762.
- Ramadhan, A.; Iyiola, K.; Alzubi, A.B. Linking absorptive capacity to project success via mediating role of customer knowledge management capability: The role of environmental complexity. *Bus. Process Manag. J.* 2024, 30, 939–962. https://doi.org/10.1108/BPMJ-07-2023-0511.
- Atik, A. Detecting income-smoothing behaviors of Turkish listed companies through empirical tests using discretionary accounting changes. Crit. Perspect. Account. 2009, 20, 591–613. https://doi.org/10.1016/j.cpa.2008.01.003.
- Aljuhmani, H.Y.; Emeagwali, O.L.; Ababneh, B. The relationships between CEOs' psychological attributes, top management team behavioral integration and firm performance. *Int. J. Organ. Theory Behav.* 2021, 24, 126–145. https://doi.org/10.1108/IJOTB-06-2020-0089.
- 63. Arend, R.J. Social and Environmental Performance at SMEs: Considering Motivations, Capabilities, and Instrumentalism. *J. Bus. Ethics* **2014**, *125*, 541–561.
- 64. Rong, B.; Wilkinson, I.F. What Do Managers' Survey Responses Mean and What Affects Them? The Case of Market Orientation and Firm Performance. *Australas. Mark. J.* 2011, *19*, 137–147. https://doi.org/10.1016/j.ausmj.2011.04.001.
- Neiroukh, S.; Emeagwali, O.L.; Aljuhmani, H.Y. Artificial intelligence capability and organizational performance: Unraveling the mediating mechanisms of decision-making processes. *Manag. Decis.* 2024, *ahead-of-print*. https://doi.org/10.1108/MD-10-2023-1946.
- Boyer, K.K.; Pagell, M. Measurement Issues in Empirical Research: Improving Measures of Operations Strategy and Advanced Manufacturing Technology. J. Oper. Manag. 2000, 18, 361–374. https://doi.org/10.1016/S0272-6963(99)00029-7.
- 67. Flynn, B.; Pagell, M.; Fugate, B. Editorial: Survey Research Design in Supply Chain Management: The Need for Evolution in Our Expectations. *J. Supply Chain Manag.* **2018**, *54*, 1–15. https://doi.org/10.1111/jscm.12161.
- 68. Armstrong, J.S.; Overton, T.S. Estimating Nonresponse Bias in Mail Surveys. J. Mark. Res. 1977, 14, 396–402. https://doi.org/10.1177/002224377701400320.
- 69. Aftab, J.; Veneziani, M.; Sarwar, H.; Abid, N. Do green practices drive business excellence in SMEs? Investigating how green entrepreneurial orientation improves firm's performance. *Total Qual. Manag. Bus. Excell.* **2024**, *35*, 529–558. https://doi.org/10.1080/14783363.2024.2315442.
- 70. Fraj, E.; Martínez, E.; Matute, J. Green marketing strategy and the firm's performance: The moderating role of environmental culture. *J. Strateg. Mark.* 2011, *19*, 339–355. https://doi.org/10.1080/0965254X.2011.581382.
- Wang, C.-H. How organizational green culture influences green performance and competitive advantage: The mediating role of green innovation. *J. Manuf. Technol. Manag.* 2019, 30, 666–683. https://doi.org/10.1108/JMTM-09-2018-0314.
- 72. Banerjee, S.B. Corporate environmentalism: The construct and its measurement. J. Bus. Res. 2002, 55, 177–191. https://doi.org/10.1016/S0148-2963(00)00135-1.
- Yavuz, O.; Uner, M.M.; Okumus, F.; Karatepe, O.M. Industry 4.0 technologies, sustainable operations practices and their impacts on sustainable performance. J. Clean. Prod. 2023, 387, 135951. https://doi.org/10.1016/j.jclepro.2023.135951.
- 74. Mousa, S.K.; Othman, M. The impact of green human resource management practices on sustainable performance in healthcare or-ganisations: A conceptual framework. *J. Clean. Prod.* **2020**, 243, 118595. https://doi.org/10.1016/j.jclepro.2019.118595.
- Nasr, E.; Emeagwali, O.L.; Aljuhmani, H.Y.; Al-Geitany, S. Destination Social Responsibility and Residents' Environmentally Responsible Behavior: Assessing the Mediating Role of Community Attachment and Involvement. *Sustainability* 2022, 14, 14153. https://doi.org/10.3390/su142114153.
- Al'Ararah, K.; Çağlar, D.; Aljuhmani, H.Y. Mitigating Job Burnout in Jordanian Public Healthcare: The Interplay between Ethical Leadership, Organizational Climate, and Role Overload. *Behav. Sci.* 2024, 14, 490. https://doi.org/10.3390/bs14060490.
- 77. Podsakoff, P.M.; MacKenzie, S.B.; Lee, J.-Y.; Podsakoff, N.P. Common method biases in behavioral research: A critical review of the literature and recommended remedies. *J. Appl. Psychol.* **2003**, *88*, 879–903. https://doi.org/10.1037/0021-9010.88.5.879.
- 78. Kock, N. Common Method Bias in PLS-SEM: A Full Collinearity Assessment Approach. Int. J. E-Collab. 2015, 11, 1–10. https://doi.org/10.4018/ijec.2015100101.
- 79. Hair, J.F.; Risher, J.J.; Sarstedt, M.; Ringle, C.M. When to use and how to report the results of PLS-SEM. *Eur. Bus. Rev.* 2019, *31*, 2–24. https://doi.org/10.1108/EBR-11-2018-0203.
- Peng, D.X.; Lai, F. Using partial least squares in operations management research: A practical guideline and summary of past research. J. Oper. Manag. 2012, 30, 467–480. https://doi.org/10.1016/j.jom.2012.06.002.
- 81. Hair, J.F.; Hult, G.T.M.; Ringle, C.; Sarstedt, M. A Primer on Partial Least Squares Structural Equations Modeling (PLS-SEM), 2nd ed.; SAGE: Thousand Oaks, CA, USA, 2017.
- 82. Fornell, C.; Larcker, D.F. Evaluating Structural Equation Models with Unobservable Variables and Measurement Error. *J. Mark. Res.* **1981**, *18*, 39–50. https://doi.org/10.1177/002224378101800104.
- Franke, G.; Sarstedt, M. Heuristics versus statistics in discriminant validity testing: A comparison of four procedures. *Internet Res.* 2019, 29, 430–447. https://doi.org/10.1108/IntR-12-2017-0515.

- Neiroukh, S.; Aljuhmani, H.Y.; Alnajdawi, S. In the Era of Emerging Technologies: DiSCB overing the Impact of Artificial Intelligence Capabilities on Timely Decision-Making and Business Performance. In Proceedings of the 2024 ASU International Conference in Emerging Technologies for Sustainability and Intelligent Systems (ICETSIS), Manama, Bahrain, 28–29 January 2024; pp. 1–6. https://doi.org/10.1109/ICETSIS61505.2024.10459356.
- Preacher, K.J.; Hayes, A.F. SPSS and SAS procedures for estimating indirect effects in simple mediation models. *Behav. Res. Methods Instrum. Comput.* 2004, *36*, 717–731. https://doi.org/10.3758/BF03206553.
- 86. Zhao, X.; Lynch, J.G., Jr.; Chen, Q. Reconsidering Baron and Kenny: Myths and Truths about Mediation Analysis. *J. Consum. Res.* 2010, *37*, 197–206. https://doi.org/10.1086/651257.
- Al-Geitany, S.; Aljuhmani, H.Y.; Emeagwali, O.L.; Nasr, E. Consumer Behavior in the Post-COVID-19 Era: The Impact of Perceived Interactivity on Behavioral Intention in the Context of Virtual Conferences. *Sustainability* 2023, 15, 8600. https://doi.org/10.3390/su15118600.
- Chin, W.W.; Marcolin, B.L.; Newsted, P.R. A Partial Least Squares Latent Variable Modeling Approach for Measuring Interaction Effects: Results from a Monte Carlo Simulation Study and an Electronic-Mail Emotion/Adoption Study. *Inf. Syst. Res.* 2003, 14, 189–217. https://doi.org/10.1287/isre.14.2.189.16018.
- 89 Henseler, J.; Chin, W.W. A Comparison of Approaches for the Analysis of Interaction Effects Between Latent Variables Using Modeling. Partial Least Squares Path Struct. Equ. Model. Multidiscip. 2010, 82-109. I. 17. https://doi.org/10.1080/10705510903439003.
- Singh, S.K.; Chen, J.; Del Giudice, M.; El-Kassar, A.N. Environmental ethics, environmental performance, and competitive advantage: Role of environmental training. *Technol. Forecast. Soc. Chang.* 2019, 146, 203–211.
- 91. Xie, X.; Huo, J.; Zou, H. Green process innovation, green product innovation, and corporate financial performance: A content analysis method. *J. Bus. Res.* 2019, 101, 697–706.
- 92. Chang, C.H. The influence of corporate environmental ethics on competitive advantage: The mediation role of green innovation. *J. Bus. Ethics* **2011**, *104*, 361–370.
- 93. Tushman, M.L.; O'Reilly, C.A., III. *Winning Through Innovation: A Practical Guide to Leading Organizational Change and Renewal;* Harvard Business School Press: Boston, MA, USA, 1997.
- 94. Kuo, F.I.; Fang, W.T.; LePage, B.A. Proactive environmental strategies in the hotel industry: Eco-innovation, green competitive advantage, and green core competence. *J. Sustain. Tour.* **2022**, *30*, 1240–1261.
- 95. Ardito, L. The influence of firm digitalization on sustainable innovation performance and the moderating role of corporate sustainability practices: An empirical investigation. *Bus. Strategy Environ.* **2023**, *32*, 5252–5272. https://doi.org/10.1002/bse.3415.
- 96. D'Attoma, I.; Ieva, M. The role of marketing strategies in achieving the environmental benefits of innovation. *J. Clean. Prod.* **2022**, 342, 130957. https://doi.org/10.1016/j.jclepro.2022.130957.
- 97. Ghisetti, C.; Montresor, S. Design and eco-innovation: Micro-evidence from the Eurobarometer survey. *Ind. Innov.* 2019, 26, 1208–1241. https://doi.org/10.1080/13662716.2018.1549475.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.